Author's response to reviews

Title: Airway, but not serum or urinary, levels of YKL-40 reflect inflammation in early cystic fibrosis lung disease.

Authors:

Emmanuelle Fantino (e.fantino@uq.edu.au)
Catherine L Gangell (c.gangell@uq.edu.au)
Dominik Hartl (dominik.hartl@med.uni-tuebingen.de)
Peter D Sly (p.sly@uq.edu.au)

Version: 3
Date: 11 February 2014

Author's response to reviews: see over
Dear Dr Cornacchia,

RE: Airway, but not circulating, YKL-40 levels reflect neutrophilic inflammation and infection in early cystic fibrosis lung disease

Thank you for consideration of the above named manuscript for publication in BMC Pulmonary Medicine.

The manuscript has been revised based on reviewer’s comments and all questions raised the reviewers have been answered.

We would like to resubmit the revised manuscript for consideration for publication.

Regards,

Peter

Professor Peter Sly
Deputy Director, Queensland Children’s Medical Research Institute
Director, Children’s Health and Environment Program, QCMRI
Professor, University of Queensland

Level 4, Foundation Building
Royal Children’s Hospital
Herston Rd
Herston, 4029, Queensland
Australia

Phone +61 7 36369690
Fax +61 7 36365578
email: p.sly@uq.edu.au